Author’s response to reviews

**Title:** Value of dual Doppler echocardiography for prediction of atrial fibrillation recurrence after radiofrequency catheter ablation

**Authors:**

Fengjiao Chen (18045623146@163.com)
Qinliang Sun (995803669@qq.com)
Hairu Li (lihairu0419@126.com)
Shaohui Qu (qshhit-2006@163.com)
Weidong Yu (156749528@qq.com)
Shuangquan Jiang (23237003@qq.com)
Jiawei Tian (jwtian2004@163.com)

**Version:** 4  **Date:** 19 Oct 2019

**Author’s response to reviews:**

Response to Editors for Manuscript ID BCAR-D-19-00477R3

1. Consent to participate

In your "Ethics approval and consent to participate" section please clearly state if the patients signed the consent forms as seen in your methods.

"All patients signed a written informed consent form prior to participating in the study."

We thank the Editor for your detailed review. We confirm that written informed consent was obtained from all participants in our study.

Now it reads “The study complied with the Declaration of Helsinki and was approved by the Ethics Committee of Harbin Medical University. All patients signed a written informed consent form prior to participating in the study” (Declarations section, line 320-323, page 15).

2. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Thank the Editor for this suggestion. I have revised the article according to the requirements. If you have any questions, please don't hesitate to contact me.